Author's response to reviews

Title: Symptom burden in community-dwelling older people with multimorbidity: a cross-sectional study

Authors:

Jeanette Eckerblad (jeanette.eckerblad@liu.se)
Kersti Theander (kersti.theander@kau.se)
Ingrid Hellstrom (ingrid.hellstrom@liu.se)
Anne Ekdahl (Anne.Ekdahl@lio.se)
Mitra Unosson (mitra.unosson@liu.se)
Ann-Britt Wirehn (Ann-Britt.Wirehn@lio.se)
Anna Milberg (anna.milberg@liu.se)
Barbro Krevers (barbro.krevers@liu.se)
Tiny Jaarsma (tiny.jaarsma@liu.se)

Version: 2 Date: 23 May 2014

Author's response to reviews: see over
<table>
<thead>
<tr>
<th>Comment</th>
<th>Changes in the text</th>
</tr>
</thead>
<tbody>
<tr>
<td>Line Numbering. Please revise your manuscript to include line and page numbers.</td>
<td>Line and page numbers are included in the manuscript</td>
</tr>
<tr>
<td>Acknowledgement. By way of a section ‘Acknowledgements’, please acknowledge anyone who contributed towards the article by making substantial contributions</td>
<td>P 15-16 row 336-342. Acknowledgements We thank all the participants and their families included in this study. We also thank the interviewers and recruiters for their assistance and support during the data collection. The first author (JE) received funding’s for this study from: the Faculty of Health sciences, Linkoping University; the county council of Ostergotland; the Signe and Olof Wallenius trust fund; Solstickan; and the Swedish Geriatric association with support from Mundipharma. The funding sources were independent and had no influence on the study.</td>
</tr>
</tbody>
</table>